Author's response to reviews

Title: Left ventricular decompression through a patent foramen ovale in a patient with hypertrophic cardiomyopathy: a case report

Authors:

Dr Giuseppe Ando' (giuseppe.ando@tiscali.it)
Fabrizio Tomai (f.tomai@tiscali.it)
Pier A Gioffre' (f.tomai@tiscali.it)

Version: 2 Date: 12 Jan 2004

Dear Dr. Picano,

enclosed please find the final version of CU manuscript ndegrees 1378586712791934 now entitled, as suggested by Reviewer 1

"Left ventricular decompression through a patent foramen ovale in a patient with hypertrophic cardiomyopathy: a case report".

A minor revision has been carried out. In particular two references, suggested by the reviewers, and a schematic new figure, as you have kindly requested, have been added.

Sincerely Yours,

Giuseppe Ando'